Author’s response to reviews

Title: Does access to healthy food vary according to socioeconomic status and to food store type? An ecologic study

Authors:
Bruna Costa (brunavlcosta@gmail.com)
Mariana Menezes (marysnut@gmail.com)
Cláudia Oliveira (claudia.dlorenzo@gmail.com)
Sueli Mingoti (suelimngt@gmail.com)
Patrícia Jaime (constant@usp.br)
Waleska Caiaffa (wcaiaffa@yahoo.com.br)
Aline Lopes (alinelopesenf@gmail.com)

Version: 4 Date: 04 May 2019

Author’s response to reviews:

We are submitting the reformulated version of the article in accordance with the suggestions of Assistant Editor. Reviewer comments and our responses are written in black and in blue for easy viewing, respectively.

• Assistant Editor Comments

1. Further Copy Edit

-- Your manuscript needs to be copyedited. We recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central has a new in house editing service. The new editing tool can provide both scientific and language editing: http://authorservices.springernature.com/.

ANSWER: Thank you for your comment. We asked American Manuscript Editors to review the manuscript. We forwarded the statement of the revision.
2. Map Attribution

-- Please state in the cover letter whether the map depicted in figure 1 is your own or taken from another source. If taken from another source please acknowledge the source in the figure legend, and if it is under copyright also state the written permission given to use and adapt it.

If the above conditions are not met the image needs to be removed. Please note the editors may request proof of permission at any time.

Should you require an alternative source you may wish to try Wikimedia Commons:

http://commons.wikimedia.org/wiki/Main_Page

ANSWER: The map depicted in figure 1 was developed by authors and exclusively for this article. Therefore, it was not published elsewhere.

3. Duplicate figures and figure legends

-- Please remove the duplicate copy of figure 1.

-- Please remove figure legends uploaded as an additional file and only include this in your figure list within your manuscript after the references.

-- Please also add a 'figure legends' heading to your figure list.

ANSWER: We removed the duplicate copy figure 1 and figure legends. We added “figure legends” after the references in the manuscript.

4. References

--Please ensure that all references are formatted according to our submission guidelines. All references, including URLs, must be numbered consecutively, in square brackets, in the order in which they are cited in the text, followed by any in tables or legends. The reference numbers must be finalized and the reference list fully formatted before submission.

Please see below:  https://bmcemergmed.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article

ANSWER: We formatted the references according to submission guidelines.
5. Clean Manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

ANSWER: We uploaded the manuscript as a final and clean version.